Author’s response to reviews

Title: An ontological approach to identifying cases of chronic kidney disease from routine primary care data: a cross-sectional study

Authors:

Nicholas Cole (ncole@doctors.org.uk)
Harshana Liyanage (h.s.liyanage@surrey.ac.uk)
Rebecca Suckling (rebecca.suckling@nhs.net)
Pauline Swift (pauline.swift1@nhs.net)
Hugh Gallagher (hugh.gallagher1@nhs.net)
Rachel Byford (r.byford@surrey.ac.uk)
John Williams (j.g.williams@surrey.ac.uk)
Shankar Kumar (shankar003@gmail.com)
Simon de Lusignan (s.lusignan@surrey.ac.uk)

Version: 1  Date: 01 Oct 2017

Author’s response to reviews:

The manuscript has been updated to include the corrections advised.

Yours sincerely,

Nicholas Cole

Corresponding author